Response to Editor and Reviewers Letter

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “Multiple inflammatory pseudotumor formation after craniopharyngioma resection via an extended nasal endoscopic approach: a case report and literature review” (ID: 88137). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our research. We have studied comments carefully and have made the correction which we hope meet with approval. The main corrections in the paper and the responses to all comments are as flowing:

Response to the editor

I have reviewed the Peer-Review Report, the full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. Before its final acceptance, the author(s) must provide the Signed Consent for Treatment Form(s) or Document(s) (患者治疗/手术同意书或病例首页). Before final acceptance, uniform presentation should be used for figures showing the same or similar contents; for example, “Figure 1 Pathological changes of atrophic gastritis after treatment. A: ...; B: ...; C: ...; D: ...; E: ...; F: ...; G: ...”. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. In order to respect and protect the author’s intellectual property rights and prevent others from misappropriating figures without the author's authorization or abusing figures without indicating the source, we will indicate the author's copyright for figures originally generated by the author, and if the author has used a figure published elsewhere or that is copyrighted, the author needs to be authorized by the previous publisher or the copyright holder and/or indicate the reference source and copyrights.
Please check and confirm whether the figures are original (i.e. generated denovo by the author(s) for this paper). If the picture is ‘original’, the author needs to add the following copyright information to the bottom right-hand side of the picture in PowerPoint (PPT): Copyright ©The Author(s) 2023.

Our Response and Action Taken
We deeply appreciate your comments, all figures are original and add Copyright ©The Author(s) 2023.
We have provided The Signed Consent for Treatment Form(s) or Document(s) (患者治疗/手术同意书或病例首页) to the journal.
According to reviewer’s comments, we have carefully modified the manuscript.

Response to Reviewer 1#
①Dear author’s, Please remove the name of your patient from the abstract.
Our Response and Action Taken
Thank you for your advising, we have removed the name of patient.
②The abstract should be reorganized. Introduction, case report and conclusion.
Our Response and Action Taken
According to your valuable comments, we have substantially modified and adjusted.
③The case should be presented in details.
Thank you for advising, according to your comments, we have added some details of case.
Our Response and Action Taken
④Please explain the novelty of your case.
Inflammatory pseudotumors is rare in the brain.
Our Response and Action Taken
⑤English edits are required.
Our Response and Action Taken
We deeply appreciate your advising, the manuscript has been edited by American Journal Experts: http://www.aje.com
Language Quality: Grade C (A great deal of language polishing)

Our Response and Action Taken

We deeply appreciate your advising, the manuscript has been polished by American Journal Experts: http://www.aje.com